Author’s response to reviews

Title: Successful treatment with anti-programmed-death-1 antibody in a relapsed natural killer/T-cell lymphoma patient with multi-line resistance: A case report

Authors:

Jianping Lai (yipingqingyun@163.com)
Peng Xu (512073602@qq.com)
Xiaoliu Jiang (923165950@qq.com)
Shan Zhou (259324509@qq.com)
Anwen Liu (awliu666@163.com)

Version: 4 Date: 13 Jul 2017

Author’s response to reviews:

Dear Rebecca Strong,

On behalf of my co-authors, we thank you very much for giving us an opportunity to revise our manuscript, we appreciate editor and reviewers very much for their positive and constructive comments and suggestions on our revised manuscript entitled “Successful treatment with anti-programmed-death-1 antibody in a relapsed natural killer/T-cell lymphoma patient with multi-line resistance: A case report”. (BCAN-D-17-00424R3).

We have studied reviewer’s comments carefully and have tried our best to revise our manuscript according to the comments. Attached please find the revised version, which we would like to submit for your kind consideration.

We would like to express our great appreciation to you and reviewers for comments on our paper. Looking forward to hearing from you.

Thank you and best regards.

Yours sincerely,

Jianping Lai
Corresponding author:
Name: Anwen Liu
E-mail: awliu666@163.com

Dear Editors and Reviewers:

Thank you for your letter and for the reviewers’ comments concerning our revised manuscript entitled “Successful treatment with anti-programmed-death-1 antibody in a relapsed natural killer/T-cell lymphoma patient with multi-line resistance: A case report”. (BCAN-D-17-00424R3). Those comments are all valuable and very helpful for revising and improving our paper, as well as the important guiding significance to our researches. We have studied comments carefully and have made correction which we hope meet with approval. The main corrections in the paper and the responds to the reviewer’s comments are as flowing:

Responds to the editor Comments:

1. Response to comment: We note that the current submission contains some textual overlap with other previously published works. Please refer to the attached iThenticate document and re-phrase these sections in your own words to ensure no overlap. If there is overlap in the Methods section, please ensure to summarize the methods and cite the source.

   In particular, please reduce the overlap in the following sections (according to the attached iThenticate document):
   - Background, Page 3, lines 31-37.
   - Background, Page 4, lines 1-20
   - Background, Page 4, lines 28-37
   - Discussion and conclusions, Page 9-10, lines 59 (page 9) to line 7 (page 10)

Response: We have re-phrased these sections according to the attached iThenticate document.

Special thanks to you for your good comments.

Yours sincerely,
Jianping Lai

Corresponding author:

Name: Anwen Liu

E-mail: awliu666@163.com